

G OPEN ACCESS

Citation: Sáez de Asteasu ML, Martínez-Velilla N, Zambom-Ferraresi F, Casas-Herrero Á, Cadore EL, Galbete A, et al. (2019) Assessing the impact of physical exercise on cognitive function in older medical patients during acute hospitalization: Secondary analysis of a randomized trial. PLoS Med 16(7): e1002852. https://doi.org/10.1371/ journal.pmed.1002852

Academic Editor: Carol Brayne, University of Cambridge, UNITED KINGDOM

Received: February 21, 2019

Accepted: June 14, 2019

Published: July 5, 2019

Copyright: © 2019 Sáez de Asteasu et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data underlying the results presented in this study can be made available on request from the Complejo Hospitalario de Navarra, by contacting Antonio García Hermoso (Antonio.garciah@unavarra.es). Public sharing of individual participant data was not included in the informed consent form of the study, but data can be made available to interested researchers who meet criteria for access to deidentified data. RESEARCH ARTICLE

Assessing the impact of physical exercise on cognitive function in older medical patients during acute hospitalization: Secondary analysis of a randomized trial

Mikel L. Sáez de Asteasu^{1,2}, Nicolás Martínez-Velilla^{1,2}, Fabricio Zambom-Ferraresi^{1,2}, Álvaro Casas-Herrero^{1,2}, Eduardo L. Cadore³, Arkaitz Galbete¹, Mikel Izquierdo^{1,2}*

1 Navarrabiomed, Complejo Hospitalario de Navarra-Universidad Pública de Navarra, IDISNA, Pamplona, Navarra, Spain, 2 CIBER of Frailty and Healthy Aging, Instituto de Salud Carlos III, Madrid, Spain, 3 Federal University of the Rio Grande of Sul, Porto Alegre, Brazil

* mikel.izquierdo@gmail.com

Abstract

Background

Acute illness requiring hospitalization frequently is a sentinel event leading to long-term disability in older people. Prolonged bed rest increases the risk of developing cognitive impairment and dementia in acutely hospitalized older adults. Exercise protocols applied during acute hospitalization can prevent functional decline in older patients, but exercise benefits on specific cognitive domains have not been previously investigated. We aimed to assess the effects of a multicomponent exercise intervention for cognitive function in older adults during acute hospitalization.

Methods and findings

We performed a secondary analysis of a single-blind randomized clinical trial (RCT) conducted from February 1, 2015, to August 30, 2017 in an Acute Care of the Elderly (ACE) unit in a tertiary public hospital in Navarre (Spain). 370 hospitalized patients (aged \geq 75 years) were randomly allocated to an exercise intervention (n = 185) or a control (n = 185) group (usual care). The intervention consisted of a multicomponent exercise training program performed during 5–7 consecutive days (2 sessions/day). The usual care group received habitual hospital care, which included physical rehabilitation when needed. The main outcomes were change in executive function from baseline to discharge, assessed with the dual-task (i.e., verbal and arithmetic) Gait Velocity Test (GVT) and the Trail Making Test Part A (TMT-A). Changes in the Mini Mental State Examination (MMSE) test and verbal fluency ability were also measured after the intervention period. The physical exercise program provided significant benefits over usual care. At discharge, the exercise group showed a mean increase of 0.1 m/s (95% confidence interval [CI], 0.07, 0.13; *p* < 0.001) in the verbal GVT and 0.1 m/s (95% CI, 0.08, 0.13; *p* < 0.001) in the arithmetic GVT over usual care group. There was an apparent improvement in the intervention group also in the TMT-A score



Funding: This study was funded by a Gobierno de Navarra project Resolución grant 2186/2014 and acknowledged with the "Beca Ortiz de Landazuri" as the best research clinical project in 2014, as well as by a research grant PI17/01814 of the Ministerio de Economía, Industria y Competitividad (ISCIII, FEDER). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: 1RM, one-repetition maximum; ACE, Acute Care of the Elderly; CAM, Confusion Assessment Method; CI, Confidence Interval; CIRS, Cumulative Illness Rating Scale; CONSORT, Consolidated Standards of Reporting Trials; EQ-VAS, visual analogue scale of the EuroQol questionnaire (EQ-5D); GDS, Yesavage Geriatric Depression Scale; GVT, Gait Velocity Test; IQR, interquartile range; MAR, missing at random; MMSE, Mini Mental State Examination; MNA, Mininutritional Assessment; Q1, First Quartile; Q3, Third Quartile; QoL, Quality of Life; RCT, randomized clinical trial; SPPB, Short Physical Performance Battery; TMT-A, Trail Making Test Part A. (-31.1 seconds; 95% Cl, -49.5, -12.7 versus -3.13 seconds; 95% Cl, -16.3, 10.2 in the control group; p < 0.001) and the MMSE score (2.10 points; 95% Cl, 1.75, 2.46 versus 0.27 points; 95% Cl, -0.08, 0.63; p < 0.001). Significant benefits were also observed in the exercise group for the verbal fluency test (mean 2.16 words; 95% Cl, 1.56, 2.74; p < 0.001) over the usual care group. The main limitations of the study were patients' difficulty in completing all the tasks at both hospital admission and discharge (e.g., 25% of older patients were unable to complete the arithmetic GVT, and 47% could not complete the TMT-A), and only old patients with relatively good functional capacity at preadmission (i.e., Barthel Index score \geq 60 points) were included in the study.

Conclusions

An individualized, multicomponent exercise training program may be an effective therapy for improving cognitive function (i.e., executive function and verbal fluency domains) in very old patients during acute hospitalization. These findings support the need for a shift from the traditional (bedrest-based) hospitalization to one that recognizes the important role of maintaining functional capacity and cognitive function in older adults, key components of intrinsic capacity.

Trial registration

ClinicalTrials.gov Identifier: NCT02300896.

Author summary

Why was this study done?

- Acute illness requiring hospitalization is often a sentinel event leading to disability.
- Acute care hospitalization increases the likelihood of developing cognitive decline in older adults.
- Physical exercise has shown to have beneficial effects on cognition both in cognitively healthy older adults and in older adults with cognitive impairment and dementia.

What did the researchers do and find?

- This study included 370 hospitalized older adults that were allocated to a physical exercise intervention group or hospital usual care group.
- An individualized multicomponent exercise intervention (i.e., progressive resistance training, balance, and gait retraining) was programmed in two daily sessions 5–7 consecutive days during hospitalization.
- The exercise program provided benefits over usual care on cognitive function in older adults.

What do these findings mean?

- Physical exercise may be an effective therapy for improving cognitive function during hospitalization.
- Only old patients with relatively good functional capacity before admission were included; thus, the generalizability of the findings is limited.
- These finding support the need for a shift from the traditional bedrest hospitalization to one that recognizes the important role of maintaining functional capacity and cognitive function in older adults.

Introduction

The provision of inpatient acute care for frail older adults has become a crucial clinical issue in our aging societies [1-3]. Acute medical illnesses and subsequent hospitalization are major events leading to disability in older people [4-6]. In addition to functional decline, acute care hospitalization increases the likelihood of developing cognitive impairment in old patients [7]. Indeed, cognitive impairment is highly prevalent in this patient group and is independently associated with multiple adverse outcomes, including functional decline, increased length of hospital stays, institutionalization, and mortality [8].

Many of the age-associated processes leading to frailty in older adults are also possibly responsible for brain aging, consecutive cognitive decline, and development of Alzheimer's disease [9]. Accordingly, frail older people are likely to be at high risk of cognitive impairment, and vice versa [9,10]. The increasing interest in the association between frailty and cognitive impairment in hospitalized older adults [11] is driving the development of innovative interventions for the prevention and management of both conditions.

Exercise and early rehabilitation protocols applied during acute hospitalization can prevent functional decline in older patients [12] and are associated with a reduced length of stay and lower costs [13]. The exercise benefits on cognitive function are not entirely clear, but previous studies support that multicomponent exercise training seems to have the most positive effects on cognition in older adults [14,15]. To the best of our knowledge, the benefits of a multicomponent exercises to attenuate cognitive impairment in acutely hospitalized older adults have not been previously investigated.

The present study is in line with the long trajectory of research that has explored new possibilities to avoid dangers of prolonged bedrest [16]. Physical exercise has shown to have beneficial effects on cognition both in cognitively healthy older adults [14,17] and in older adults with cognitive impairment or dementia [18,19]. Thus, the main purpose of our study was to assess the effects of a multicomponent exercise intervention for cognitive function in older adults during acute hospitalization. Considering the strong link between physical exercise and cognition, our hypothesis was that multicomponent exercise intervention would maintain or even improve cognitive function compared to usual care in these patients.

Methods

Design

The study is a secondary analysis of a randomized clinical trial (RCT) (NCT02300896) [12,20]. This study is reported as per the Consolidated Standards of Reporting Trials (CONSORT) guideline (S1 CONSORT Checklist). Differing with the previous analysis, in this study, specific cognitive domains described in the study protocol [20] (i.e., executive function, verbal fluency ability) were analyzed, and other cognitive tasks such as dual tasks were included. It was conducted in the Acute Care of the Elderly (ACE) unit of the Department of Geriatrics in a tertiary public hospital (Complejo Hospitalario de Navarra, Spain). This department has 35 allocated beds, and its staff is composed of 8 geriatricians (distributed in the ACE unit, orthogeriatrics, and outpatient consultations). Admissions in the ACE unit derive mainly from the Accident and Emergency Department, with heart failure and pulmonary and infectious diseases being the main causes of admissions.

Acutely hospitalized patients who met inclusion criteria were randomly assigned to the intervention or control (usual care) group within the first 48 hours of admission. Usual care is offered to patients by the geriatricians of our department and consists of standard physiotherapy focused on walking exercises for restoring the functionality conditioned by potentially reversible pathologies. A formal exercise prescription was not provided at study entry, and patients were instructed to continue with the current activity practices through the duration of the study. The study followed the principles of the Declaration of Helsinki and was approved by the Complejo Hospitalario de Navarra Research Ethics Committee. All patients or their legal representatives provided written consent.

Participants and randomization

All of the patients admitted to the ACE unit were evaluated by geriatricians. We focused on a particularly vulnerable population, but also one with a level of functional and cognitive capacity high enough to allow them to perform the physical exercise protocol. A trained research assistant conducted a screening interview to determine whether potentially eligible patients met the following inclusion criteria: age \geq 75 years, Barthel Index score \geq 60 points, and able to ambulate (with/without assistance) and to communicate and collaborate with the research team. Exclusion criteria included expected length of stay <6 days, very severe cognitive decline (i.e., Global Deterioration Scale score = 7), terminal illness, uncontrolled arrhythmias, acute pulmonary embolism and myocardial infarction, or extremity bone fracture in the past 3 months.

After the baseline assessment was performed, participants were randomly assigned following a 1:1 ratio without restrictions. The simple randomization sequence was generated by a statistician not involve in the RCT using an online system (www.randomizer.org) to allocate 185 patients in the exercise group (intervention group) and 185 patients in the usual care group (control group). Assessment staff was blinded to the main study design and group allocation. It was not possible to blind the participants, so they were explicitly informed and reminded not to discuss their randomization assignment with the assessment staff.

Intervention

The usual care group received habitual hospital care, which included physical rehabilitation when needed. For the intervention group, exercise training was programmed in two daily sessions (morning and evening) of 20 minutes duration during 5–7 consecutive days (including

weekends) supervised by a qualified fitness specialist. Adherence to the exercise intervention program was recorded in a daily register. A session was considered completed when \geq 90% of the programmed exercises were successfully undertaken.

Each session was performed in a room equipped ad hoc in the ACE unit. Exercises were adapted from the "Vivifrail" multicomponent physical exercise program to prevent weakness and falls [21]. The morning sessions included individualized progressive resistance, balance, and walking-training exercises and were supervised by a physiotherapist (M.L.S.de.A) or a researcher (F.Z.F) with a PhD background in exercise physiology. The resistance exercises were tailored to the individual's functional capacity using variable resistance training machines (Matrix; Johnson Health Tech, Ibérica, S.L., Torrejón de Ardoz, Spain and Exercycle S.L.; BHGroup, Vitoria, Spain) aiming at 2-3 sets of 8-10 repetitions with a load equivalent to 30%-60% of the estimated one-repetition maximum (1RM). Participants performed three exercises involving mainly lower-limb muscles (squats rising from a chair, leg press, and bilateral knee extension) and one involving the upper-body musculature (seated bench "chest" press). They were instructed to perform the exercises at a high speed to optimize muscle power output, and care was taken to ensure proper exercise execution. Balance and gait-retraining exercises gradually progressed in difficulty and included the following: semitandem foot standing, line walking, stepping practice, walking with small obstacles, proprioceptive exercises on unstable surfaces (foam pad sequence), altering the base of support, and weight transfer from one leg to the other. The evening session consisted of functional unsupervised exercises using light loads (0.5-1 kg anklets and hand-grip ball), such as knee extension/flexion, hip abduction, and daily walking in the corridor of the ACE unit with a duration based on the clinical physical exercise guide "Vivifrail" [21].

When the clinician in charge of the patient considered that the hemodynamic situation was acceptable and the patient could collaborate, the following endpoints were assessed, and the intervention was started. Endpoints were also assessed on the day of discharge.

Endpoints

The primary endpoint was change in executive function during hospitalization (i.e., from admission to discharge) as assessed with the 6-meter dual-task Gait Velocity Test (GVT). Secondary endpoints were the Mini Mental State Examination (MMSE) test, the Trail Making Test Part A (TMT-A), and the verbal fluency test. The MMSE endpoint was included in the main analysis of the RCT [12].

6-meter dual-task GVT

Patients were instructed to walk at their self-selected usual pace on a smooth, horizontal walkway. Two different dual-task gait tests were performed, the arithmetic GVT and the verbal GVT, in which gait velocity was measured while the patients counted backward aloud from 100 down to 1 or named animals aloud, respectively [22,23]. The cognitive score was measured by counting the number of animals named (verbal dual task) or by counting the numbers that were stated (arithmetic dual task) and the errors in each task.

MMSE

The MMSE test [24] is the most utilized screening instrument of cognitive decline [25]. The instrument assesses domains of orientation, memory, attention, language, and visuospatial ability. The MMSE is scored out of 30 points, with scores \leq 23 points indicative of likely cognitive impairment.

TMT-A

The TMT-A is used as an indicator of visual scanning, graphomotor speed, and executive function. The patients were asked to connect randomly arranged circles containing numbers from 1 to 25 following the number sequence and to do it as quickly as possible [26].

Verbal fluency test

The patient had to say as many words as possible starting with the letter F in one minute [27].

Statistical analysis

Analyses were performed by "intention-to-treat" principles. After analyzing missing-data patients in both groups and comparing them with the non-missing-data patients, a missing at random (MAR) mechanism was assumed.

Between-group comparisons of continuous variables were conducted using linear mixed models. Time was treated as a categorical variable. The models included group, time, and group by time interaction as fixed effects and participants as random effect. For each group, data are expressed as change from baseline (admission) to discharge, determined by the time coefficients (95% confidence interval [CI]) of the model. The conclusions about effectiveness of exercise intervention were based on between-group comparisons of change in cognitive function from baseline (beginning of the intervention) to hospital discharge, as assessed with the MMSE, dual-task GVT (including both verbal and arithmetic task conditions), TMT-A, and verbal fluency test and determined by the time by group interaction coefficients of the model. Between-group comparisons of errors during the dual-task GVTs were analyzed using the Poisson mixed model because of the asymmetric distribution of the endpoint.

Using the χ^2 test for linear trend, we also compared the proportion of patients in each group showing an improvement, no change, or worsening at discharge at compared with base-line on the dual-task GVTs.

Normality of data was checked graphically and through the Kolmogorov–Smirnov test. The residuals were also checked graphically, and no noticeable deviation from normality was observed. All comparisons were two-sided, with a significance level of 0.05. Statistical analysis was carried out using IBM-SPSS v20 software (SPSS Inc., Chicago, IL, USA).

Results

The study flow diagram is shown in Fig 1. No significant differences were found between groups at baseline for demographic and clinical characteristics for study endpoints (Table 1). Of the 370 patients included in the analyses, 209 were women (56.5%); the mean age was 87.3 (4.9) years (range 75–101 years), with 130 patients (35.1%) being nonagenarians). The median length of hospital stay was 8 days in both groups (interquartile range [IQR], 4 and 4 days, respectively). The mean number of intervention days for each patient was 5.3 ± 0.5 days, with most training days being consecutive (97%). The number of completed morning and evening sessions per patient averaged 5 ± 1 and 4 ± 1 , respectively. Mean adherence to the intervention was 97% (95% CI, 95.67, 98.36) for the morning sessions (i.e., 806 successfully completed sessions of 841 total possible sessions) and 85% (95% CI, 79.70, 89.40) in the evening sessions (574 of 688). No adverse effects or falls associated with the prescribed exercises were recorded, and no patient had to interrupt the intervention or had their hospital stay modified because of it.

The primary analysis showed that the physical exercise seems to provide a significant benefit over usual care. Differences between the treatment groups revealed a significant





https://doi.org/10.1371/journal.pmed.1002852.g001

intervention effect for both dual-task GVTs. The percentage distribution of patients with improvements on the verbal GVT (47.6% versus 81.7%) or arithmetic GVT (48.7% versus 88.5%) from admission to discharge significantly differed between the two groups, indicating a beneficial exercise intervention effect for both endpoints (all p < 0.001 with χ^2 test, Fig 2A and 2B). At discharge, the exercise group showed an increase of 0.1 m/s (95% CI, 0.07, 0.13 m/s; p < 0.001) on the verbal GVT and 0.1 m/s (95% CI, 0.08, 0.13 m/s; p < 0.001) on the verbal GVT and 0.1 m/s (95% CI, 0.08, 0.13 m/s; p < 0.001) on the arithmetic GVT over the usual care group (Table 2, Fig 2C and 2D). Furthermore, significant enhancements were found in the intervention group in the errors made during the arithmetic GVT (0.48 errors; 95% CI, 0.34, 0.67 words) over the control group (p < 0.001, Table 2).

Variable	Control Group (n = 185)	Intervention Group (n = 185)	
Demographic data			
Age, years	87.1 (5.2)	87.6 (4.6)	
Women, N (%)	109 (59%)	100 (54%)	
Body mass index, kg/m ²	26.9 (4.9)	27.1 (4.4)	
Clinical data			
Barthel Index, score	83 (17)	84 (17)	
CIRS score, median (IQR)	12 (5)	13 (5)	
MNA score, median (IQR)	24 (4)	24 (4)	
1RM leg press, kg	62 (31)	57 (25)	
1RM chest press, kg	25 (12)	24 (11)	
1RM knee extension, kg	41 (14)	39 (13)	
GDS, score	3.6 (2.9)	4.0 (2.4)	
QoL (EQ-VAS), score	60 (21)	58 (22)	
Delirium (CAM, %)	12%	17%	
Endpoint measures			
Verbal GVT, m/s	0.4 (0.2)	0.4 (0.2)	
Arithmetic GVT, m/s	0.4 (0.2)	0.4 (0.2)	
MMSE, score	23 (4)	22 (5)	
TMT-A, seconds	162.9 (97.0)	166.5 (125.4)	
Verbal fluency test, score	7.2 (4.2)	6.3 (3.8)	
Admission reason, N (%)			
Cardiovascular	67 (36)	65 (35)	
Infectious	33 (18)	33 (18)	
Pulmonary	20 (11)	28 (15)	
Gastrointestinal	17 (9)	20 (11)	
Neurological	9 (5)	9 (5)	
Other	39 (21)	30 (16)	

Table 1. Baseline characteristics of the participants.

Data are mean (SD) unless otherwise stated. No statistically significant differences were found between groups (all p > 0.05).

Abbreviations: CAM, Confusion Assessment Method; CIRS, Cumulative Illness Rating Scale; EQ-VAS, visual analogue scale of the EuroQol questionnaire (EQ-5D); GDS, Yesavage Geriatric Depression Scale; GVT, Gait Velocity Test; IQR, interquartile range; MMSE, Mini Mental State Examination; MNA, Mini-nutritional Assessment; QoL, quality of life; SPPB, Short Physical Performance Battery; TMT-A, Trail Making Test Part A; 1RM, one-repetition maximum.

https://doi.org/10.1371/journal.pmed.1002852.t001

Considering the global cognitive function, the intervention group showed improvements at discharge in the MMSE test of 2.10 points (95% CI, 1.75, 2.46 points), whereas no such trend was found in the control group (0.27 points; 95% CI, -0.08, 0.63 points) (p < 0.001) (Table 2 and Fig 3A).

For the executive function, the exercise group showed an improvement in the TMT-A, reducing the time to complete the task by 31.1 seconds at discharge (95% CI, -49.5 to -12.7 seconds; p < 0.001) over the control group (Table 2 and Fig 3B).

Finally, acute hospitalization per se led to a significant impairment in patient verbal fluency ability (i.e., mean change from baseline to discharge of -0.30 words [95% CI, -0.72, 0.12 words]), whereas the exercise intervention improved this cognitive domain (1.85 words; 95% CI, 1.44, 2.27 words) (p < 0.001) (Table 2 and Fig 3C).



Fig 2. Changes from baseline to discharge (A and B) and within-group punctuation change distribution (C and D). Dual-task GVT changes: "better" indicates an improvement of more than 0.1 m/s, "slightly better" indicates an improvement between 0.001 and 0.1 m/s, "unchanged" indicates no difference, "slightly worse" indicates a decline between 0.001 and 0.1 m/s, and "worse" indicates a decline of more than 0.1 m/s. The proportion of patients showing overall improvement and worsening in the dual-task GVTs was significantly higher and lower, respectively, in the intervention than in the control group (all p < 0.001 with χ^2 test). In the violin plots, the horizontal dotted lines indicate Q1 and Q3, and the horizontal dashed line within the violin, the median. GVT, Gait Velocity Test; Q1, First Quartile; Q3, Third Quartile.

https://doi.org/10.1371/journal.pmed.1002852.g002

Discussion

This secondary analysis of the RCT suggests that an individualized exercise intervention, delivered over a mean of 5 days, may provide benefits over usual care in acutely hospitalized older adults and may help reverse the cognitive impairment often associated with this patient group. To our knowledge, this is the first study to point towards the beneficial effects of a multicomponent intervention, including low-intensity resistance training exercises on specific cognitive domains such as executive function and verbal fluency, in hospitalized patients of advanced age.

Older patients admitted to the hospital are at risk of experiencing negative consequences following hospitalization, including functional decline and, frequently, long-term disability

Endpoints	Control Group	Exercise	Between-Group Difference (95% CI)	<i>p</i> -Value Between Groups
Verbal GVT		1		
Velocity, m/s	0.002 (-0.018, 0.022)	0.10 (0.08, 0.12)	0.10 (0.07, 0.13)	<0.001
Correct answers, score	0.01 (-0.36, 0.38)	0.41 (0.04, 0.79)	0.41 (-0.12, 0.93)	0.133
Errors, score*	2.03 (0.64, 7.61)	0.32 (0.016, 2.41)	0.16 (0.01, 1.65)	0.157
Arithmetic GVT				
Velocity, m/s	0.009 (-0.009, 0.029)	0.11 (0.10, 0.13)	0.10 (0.08, 0.13)	<0.001
Correct answers, score	0.12 (-0.57, 0.81)	0.18 (-0.52, 0.88)	0.06 (-0.92, 1.05)	0.901
Errors, score*	1.16 (0.92, 1.45)	0.55 (0.42, 0.69)	0.48 (0.34, 0.67)	<0.001
MMSE, score	0.27 (-0.08, 0.63)	2.10 (1.75, 2.46)	1.83 (1.32, 2.33)	<0.001
TMT-A, seconds	-3.13 (-16.3, 10.2)	-34.2 (-47.1, -21.3)	-31.1 (-49.5, -12.7)	< 0.001
Verbal fluency test				
Correct answers, score	-0.30 (-0.72, 0.12)	1.85 (1.44, 2.27)	2.16 (1.56, 2.74)	<0.001
Errors, score*	1.11 (0.75, 1.63)	0.66 (0.43, 0.99)	0.58 (0.33, 1.05)	0.076

Table 2. Results of study endpoints by group.

Data in each group are expressed as change from baseline (admission) to discharge (mean and 95% CI). A total of 247 patients (79% of 310 patients who completed the intervention) were able to perform the verbal GVT at admission and discharge; 231 patients (75%) the arithmetic GVT; 292 older adults (94%) the MMSE test; 162 patients (53%) the TMT-A; and 289 patients (93%) the verbal fluency test.

*Poisson mixed model. Effects are rate ratios: within-group effects are time coefficients in the model, and between-group effects are group-time interaction coefficients. Abbreviations: CI, confidence interval; GVT, Gait Velocity Test; MMSE, Mini Mental State Examination; TMT-A, Trail Making Test Part A.

https://doi.org/10.1371/journal.pmed.1002852.t002

[5,6]. Research has suggested that hospitalization in older adults per se is associated not only with functional adverse outcomes but also with the development of cognitive decline and an increased risk of dementia [7]. Moreover, cognitively impaired older patients are at even greater risk of hazards of hospital stay as compared to patients with no cognitive decline [28]. Our findings reveal that more than one-half of the control group showed worsened gait performance in both the dual-task GVTs at discharge, whereas the exercise intervention reversed this trend. We also observed an improvement in the verbal fluency ability after the exercise





https://doi.org/10.1371/journal.pmed.1002852.g003

intervention, with the opposite response found in the usual care group. Surprisingly, shortterm hospitalization did not impact dramatically on some cognitive tasks, such as the MMSE and TMT-A, although significant differences were observed between groups at discharge. The poor health status of the hospitalized old patients upon admission and the comprehensive and multidisciplinary protocols already established in the ACE unit could influence the preservation of some cognitive domains.

Acute hospital admissions play a key role in the disabling process in older adults, and physical exercise seems to be an effective therapy to prevent nosocomial disability, which is usually linked to poor mobility during hospitalization [29]. Recent evidence has demonstrated that specific in-hospital exercises could provide significant benefits over usual care and could help to reverse the functional decline associated with acute hospitalization in older adults [12]. Although potential benefits of physical exercise on functional capacity are well established, the effects of tailored multicomponent exercise intervention on specific cognitive domains including executive function and verbal fluency are not clear in acutely hospitalized older patients. In agreement with previous studies [14,15], our findings support that multicomponent exercise training may produce the most positive effects on cognitive function in older adults. The inclusion of progressive low-intensity resistance training as a component of the exercise training protocol could be the reason for cognitive gains in the intervention group in specific executive tasks (i.e., both dual-task GVTs and TMT-A). An emerging theory to explain these cognitive benefits is that resistance training increases the production of several growth factors, such as brain-derived neurotrophic factor and insulin-like growth factor 1 [30]. Previous evidence has suggested that gait performance is closely related to cognitive function, in particular executive function, and impaired executive function has been associated with decreased gait velocity, increased risk of falls, and decreased performance on complex motor tasks in older adults [31,32]. Thus, our results indicate that, despite its short duration, an exercise training approach can be effective in improving the executive function (measured by dual-task GVT) during hospitalization in very old patients.

The present study is in line with the recently published World Health Organization (WHO) Clinical Consortium of Healthy Aging, which highlights the importance of maintaining individuals' intrinsic capacity for the preservation of autonomy and independence in essential everyday activities [33]. Our findings suggest that a multicomponent exercise program with special emphasis on muscle-power training may help mitigate the trajectory towards frailty or disability in acutely hospitalized older adults and appeared to improve cognitive function, a key component of intrinsic capacity. These data suggest that, in accordance with the WHO framework, exercise prescription should be considered as part of frontline treatment to prevent hospital-acquired iatrogenic disability. Future RCTs should also consider the inclusion of multidomain interventions, in which exercise training is combined with other treatments such as cognitive training and social enrichment, in this population to optimize cognitive performance and prevent cognitive impairment.

Our study has several strengths. We focused on a particularly vulnerable population of advanced age (overall mean 87.3 years; range 75–101 years, with 130 patients [35.1%] being nonagenarians) to develop an innovative exercise intervention of a few days (i.e., 5 ± 1 and 4 ± 1 morning and evening sessions, respectively) in acute settings. Also, patients with multiple comorbidities (mean [SD] of 9 [6] comorbidities) and mild dementia/cognitive impairment were included in the RCT (routinely excluded from exercise studies). Considering the exercise training protocol, a daily individualized adjustment of loads was performed to optimize exercise benefits and prevent iatrogenic nosocomial disability. In order to minimize potential bias, the researchers were unaware of patient test scores at admission when retesting at discharge. Our study, nevertheless, has some limitations, including patients' difficulty in completing all

the tasks at both hospital admission and discharge. Notably, 25% of the older patients were unable to perform the arithmetic GVT, mainly because they did not receive primary education, and 47% of the participants could not complete the TMT-A because of visual impairment. Thus, the missing data were not possible to collect because of the characteristics of the study population (octogenarians and nonagenarians with multiple geriatric syndromes). However, the data missingness was not differential between the intervention and control group, and it seems not to affect to the between-group differences. The generalizability of our results is limited due to the inclusion of a selected population with relatively good functional capacity at preadmission (i.e., Barthel Index score ≥ 60 points), excluding those older adults with severe dementia, with unstable hemodynamic condition, or not able to walk at admission. Physical exercise during acute hospitalization may be beneficial for avoiding bedrest complications, but further research is needed in other hospitalized older medical populations, and it would be interesting to replicate our findings in a multicenter study. Additionally, it was not possible to blind physical therapists in charge of supervising the intervention, patients, or families to the group assignment of a patient given the nature of the intervention. Finally, delirium data were not included in this secondary analysis because they were added in the main analysis of the RCT [12].

Our findings point to several future directions for research. In general, physical exercise intervention seems to provide benefits on cognition over usual care in acutely hospitalized older adults. Future studies should analyze the individual response of participants to usual care (control group) and to physical exercise (intervention group) for cognitive function to prescribe more individualized treatments and optimize the results. Additionally, the role of exercise training on cognition at follow-up (i.e., post-discharge) needs to be further explored in trials.

In this study, we evaluated the effects of an individualized, multicomponent exercise program on cognitive function in hospitalized older adults. The physical exercise intervention appears to be an effective therapy for improving cognitive function (i.e., executive function and verbal fluency domains) in very old patients during acute hospitalization. These findings support the need for a shift from the traditional (bedrest-based) hospitalization to one that recognizes the important role of maintaining functional capacity and cognitive function in older adults, key components of intrinsic capacity.

Supporting information

S1 CONSORT Checklist. CONSORT 2010 checklist. (DOC)

S1 Data. (XLSX)

Author Contributions

Conceptualization: Mikel L. Sáez de Asteasu, Nicolás Martínez-Velilla, Fabricio Zambom-Ferraresi, Álvaro Casas-Herrero, Arkaitz Galbete, Mikel Izquierdo.

Data curation: Arkaitz Galbete.

Formal analysis: Mikel L. Sáez de Asteasu, Nicolás Martínez-Velilla.

Funding acquisition: Nicolás Martínez-Velilla, Mikel Izquierdo.

Investigation: Mikel L. Sáez de Asteasu, Nicolás Martínez-Velilla, Fabricio Zambom-Ferraresi, Álvaro Casas-Herrero, Eduardo L. Cadore, Arkaitz Galbete, Mikel Izquierdo. Methodology: Nicolás Martínez-Velilla, Álvaro Casas-Herrero, Arkaitz Galbete, Mikel Izquierdo.

Resources: Mikel Izquierdo.

Software: Fabricio Zambom-Ferraresi, Eduardo L. Cadore.

Supervision: Nicolás Martínez-Velilla, Mikel Izquierdo.

Validation: Álvaro Casas-Herrero.

Visualization: Mikel L. Sáez de Asteasu.

Writing – original draft: Mikel L. Sáez de Asteasu, Nicolás Martínez-Velilla, Mikel Izquierdo.

Writing – review & editing: Mikel L. Sáez de Asteasu, Nicolás Martínez-Velilla, Fabricio Zambom-Ferraresi, Álvaro Casas-Herrero, Eduardo L. Cadore, Arkaitz Galbete, Mikel Izquierdo.

References

- Gilbert T, Neuburger J, Kraindler J, Keeble E, Smith P, Ariti C, et al. Development and validation of a Hospital Frailty Risk Score focusing on older people in acute care settings using electronic hospital records: an observational study. Lancet 2018; 391(10132): 1775–1782. https://doi.org/10.1016/S0140-6736(18)30668-8 PMID: 29706364
- Rechel B, Grundy E, Robine JM, Cylus J, Mackenbach JP, Knai C, et al. Ageing in the European Union. Lancet 2013; 381(9874): 1312–1322. https://doi.org/10.1016/S0140-6736(12)62087-X PMID: 23541057
- 3. Spillman BC, Lubitz J. The effect of longevity on spending for acute and long-term care. N Engl J Med 2000; 342(19):1409–1415. https://doi.org/10.1056/NEJM200005113421906 PMID: 10805827
- Covinsky KE, Palmer RM, Fortinsky RH, Counsell SR, Stewart AL, Kresevic D, et al. Loss of independence in activities of daily living in older adults hospitalized with medical illnesses: increased vulnerability with age. J Am Geriatr Soc 2003; 51(4): 451–458. PMID: 12657063
- Gill TM, Allore HG, Holford TR, Guo Z. Hospitalization, restricted activity, and the development of disability among older persons. JAMA 2004; 292(17): 2115–2124. <u>https://doi.org/10.1001/jama.292.17</u>. 2115 PMID: 15523072
- Gill TM, Gahbauer EA, Han L, Allore HG. The role of intervening hospital admissions on trajectories of disability in the last year of life: prospective cohort study of older people. BMJ 2015; 350: h2361. <u>https:// doi.org/10.1136/bmj.h2361</u> PMID: 25995357
- Ehlenbach WJ, Hough CL, Crane PK, Haneuse SJPA, Carson SS, Curtis JR, et al. Association between acute care and critical illness hospitalization and cognitive function in older adults. JAMA 2010; 303(8): 763–770. https://doi.org/10.1001/jama.2010.167 PMID: 20179286
- Lucke JA, van der Mast RC, De GJ, Heim N, De Groot B, Mooijaart SP, et al. The Six-Item Cognitive Impairment Test Is Associated with Adverse Outcomes in Acutely Hospitalized Older Patients: A Prospective Cohort Study. Dement Geriatr Cogn Dis Extra 2018; 8: 259–267. <u>https://doi.org/10.1159/</u> 000490240 PMID: 30140275
- Wallace LMK, Theou O, Godin J, Andrew MK, Bennett DA, Rockwood K. Investigation of frailty as a moderator of the relationship between neuropathology and dementia in Alzheimer's disease: a crosssectional analysis of data from the Rush Memory and Aging Project. Lancet Neurol 2019; 18(2): 177– 184. https://doi.org/10.1016/S1474-4422(18)30371-5 PMID: 30663607
- Robertson DA, Savva GM, Kenny RA. Frailty and cognitive impairment—a review of the evidence and causal mechanisms. Ageing Res Rev 2013; 12(4): 840–851. https://doi.org/10.1016/j.arr.2013.06.004 PMID: 23831959
- Sands LP, Yaffe K, Covinsky K, Chren MM, Counsell S, Palmer R, et al. Cognitive screening predicts magnitude of functional recovery from admission to 3 months after discharge in hospitalized elders. J Gerontol A Biol Sci Med Sci 2003; 58(1): 37–45. https://doi.org/10.1093/gerona/58.1.m37 PMID: 12560409
- Martinez-Velilla N, Casas-Herrero A, Zambom-Ferraresi F, Saez de Asteasu ML, Lucia A, Galbete A, et al. Effect of Exercise Intervention on Functional Decline in Very Elderly Patients During Acute Hospitalization: A Randomized Clinical Trial. JAMA Intern Med 2019; 179(1): 28–36.
- de Morton NA, Keating JL, Jeffs K. Exercise for acutely hospitalised older medical patients. Cochrane Database Syst Rev 2007;(1):CD005955. https://doi.org/10.1002/14651858.CD005955.pub2 PMID: 17253572

- Colcombe S, Kramer AF. Fitness effects on the cognitive function of older adults: a meta-analytic study. Psychol Sci 2003; 14(2): 125–130. https://doi.org/10.1111/1467-9280.t01-1-01430 PMID: 12661673
- Saez de Asteasu ML, Martinez-Velilla N, Zambom-Ferraresi F, Casas-Herrero A, Izquierdo M. Role of physical exercise on cognitive function in healthy older adults: A systematic review of randomized clinical trials. Ageing Res Rev 2017; 37: 117–134. <u>https://doi.org/10.1016/j.arr.2017.05.007</u> PMID: 28587957
- 16. ASHER RA. The dangers of going to bed. Br Med J 1947; 2(4536): 967. https://doi.org/10.1136/bmj.2. 4536.967 PMID: 18897489
- Etnier JL, Salazar W, Landers DM, Petruzello SJ, Han M, Nowell P. The influence of physical fitness and exercise upon cognitive functioning: A meta-analysis. J Sport Exerc Psychol 1997; 19(3): 249–277.
- Eggermont L, Swaab D, Luiten P, Scherder E. Exercise, cognition and Alzheimer's disease: more is not necessarily better. Neurosci Biobehav Rev 2006; 30(4): 562–575. https://doi.org/10.1016/j.neubiorev. 2005.10.004 PMID: 16359729
- Heyn P, Abreu BC, Ottenbacher KJ. The effects of exercise training on elderly persons with cognitive impairment and dementia: a meta-analysis. Arch Phys Med Rehabil 2004; 85(10): 1694–1704. PMID: 15468033
- 20. Martinez-Velilla N, Casas-Herrero A, Zambom-Ferraresi F, Suarez N, Alonso-Renedo J, Contín KC, et al. Functional and cognitive impairment prevention through early physical activity for geriatric hospitalized patients: study protocol for a randomized controlled trial. BMC Geriatr 2015; 15(1): 112.
- Izquierdo M, Casas-Herrero A, Zambom-Ferraresi F, Martínez-Velilla N, Alonso-Bouzón C, Rodríguez-Mañas L. Multicomponent Physical Exercise program VIVIFRAIL. 2017 [cited 2019 Feb 1]. <u>http://www.vivifrail.com/images/recursos/VIVIFRAIL-ENG-Interactivo.pdf</u>.
- Beauchet O, Annweiler C, Dubost V, Allali G, Kressig RW, Bridenbaugh S, et al. Stops walking when talking: a predictor of falls in older adults? Eur J Neurol 2009; 16(7): 786–795. https://doi.org/10.1111/j. 1468-1331.2009.02612.x PMID: 19473368
- Li KZH, Bherer L, Mirelman A, Maidan I, Hausdorff JM. Cognitive Involvement in Balance, Gait and Dual-Tasking in Aging: A Focused Review From a Neuroscience of Aging Perspective. Front Neurol 2018; 9: 913. https://doi.org/10.3389/fneur.2018.00913 PMID: 30425679
- Folstein MF, Folstein SE, McHugh PR. "Mini-mental state". A practical method for grading the cognitive state of patients for the clinician. J Psychiatr Res 1975; 12(3): 189–198. PMID: 1202204
- Brodaty H, Connors MH, Loy C, Teixeira-Pinto A, Stocks N, Gunn J, et al. Screening for Dementia in Primary Care: A Comparison of the GPCOG and the MMSE. Dement Geriatr Cogn Disord 2016; 42(5–6): 323–330. https://doi.org/10.1159/000450992 PMID: 27811463
- Llinas-Regla J, Vilalta-Franch J, Lopez-Pousa S, Calvo-Perxas L, Torrents RD, Garre-Olmo J. The Trail Making Test. Assessment 2017; 24(2): 183–196. <u>https://doi.org/10.1177/1073191115602552</u> PMID: 26318386
- 27. Wysokinski A, Zboralski K, Orzechowska A, Galecki P, Florkowski A, Talarowska M. Normalization of the Verbal Fluency Test on the basis of results for healthy subjects, patients with schizophrenia, patients with organic lesions of the chronic nervous system and patients with type 1 and 2 diabetes. Arch Med Sci 2010; 6(3): 438–446. https://doi.org/10.5114/aoms.2010.14268 PMID: 22371783
- Provencher V, Sirois MJ, Ouellet MC, Candem S, Neveu X, Allain-Boulé N, et al. Decline in activities of daily living after a visit to a Canadian emergency department for minor injuries in independent older adults: are frail older adults with cognitive impairment at greater risk? J Am Geriatr Soc 2015; 63(5): 860–868. https://doi.org/10.1111/jgs.13389 PMID: 25989564
- Karlsen A, Loeb MR, Andersen KB, Joergensen KJ, Scheel FU, Turtumoeygard IF, et al. Improved Functional Performance in Geriatric Patients During Hospital Stay. Am J Phys Med Rehabil 2017; 96 (5): e78–e84. https://doi.org/10.1097/PHM.000000000000671 PMID: 28045706
- Kramer AF, Erickson KI. Capitalizing on cortical plasticity: influence of physical activity on cognition and brain function. Trends Cogn Sci 2007; 11(8): 342–348. https://doi.org/10.1016/j.tics.2007.06.009 PMID: 17629545
- Montero-Odasso M, Muir SW, Speechley M. Dual-task complexity affects gait in people with mild cognitive impairment: the interplay between gait variability, dual tasking, and risk of falls. Arch Phys Med Rehabil 2012; 93(2): 293–299. https://doi.org/10.1016/j.apmr.2011.08.026 PMID: 22289240
- Persad CC, Jones JL, Ashton-Miller JA, Alexander NB, Giordani B. Executive function and gait in older adults with cognitive impairment. J Gerontol A Biol Sci Med Sci 2008; 63(12): 1350–1355. <u>https://doi.org/10.1093/gerona/63.12.1350 PMID: 19126848</u>
- WHO Clinical Consortium on Healthy Ageing 2017 –report of consortium meeting 21 and 22 November 2017 in Geneva, Switzerland. Geneva: World Health Organization; 2018.